

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

The distribution of fitness effects of spontaneous mutations in Chlamydomonas reinhardtii inferred using frequency changes under experimental evolution

Citation for published version:

Böndel, KB, Samuels, T, Craig, RJ, Ness, RW, Colegrave, N, Keightley, PD & Slotte, T (ed.) 2022, 'The distribution of fitness effects of spontaneous mutations in Chlamydomonas reinhardtii inferred using frequency changes under experimental evolution', *PLoS Genetics*, vol. 18, no. 6, e1009840. https://doi.org/10.1371/journal.pgen.1009840

Digital Object Identifier (DOI):

10.1371/journal.pgen.1009840

Link:

Link to publication record in Edinburgh Research Explorer

Document Version: Publisher's PDF, also known as Version of record

Published In: PLoS Genetics

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





Citation: Böndel KB, Samuels T, Craig RJ, Ness RW, Colegrave N, Keightley PD (2022) The distribution of fitness effects of spontaneous mutations in *Chlamydomonas reinhardtii* inferred using frequency changes under experimental evolution. PLoS Genet 18(6): e1009840. https:// doi.org/10.1371/journal.pgen.1009840

Editor: Tanja Slotte, Stockholm University, SWEDEN

Received: September 24, 2021

Accepted: April 13, 2022

Published: June 15, 2022

Copyright: © 2022 Böndel et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Sequencing reads are available at http://www.ncbi.nlm.nih.gov/bioproject/ 843600. Programs, scripts and processed data at https://sourceforge.net/projects/dfe-inchlamydomonas/.

Funding: PDK received funding from the European Research Council under the European Union's Horizon 2020 Research And Innovation Programme (Grant Agreement no. 694212, https:// erc.europa.eu/). The funders had no role in study RESEARCH ARTICLE

The distribution of fitness effects of spontaneous mutations in *Chlamydomonas reinhardtii* inferred using frequency changes under experimental evolution

Katharina B. Böndel^{1*}, Toby Samuels¹, Rory J. Craig¹, Rob W. Ness², Nick Colegrave¹, Peter D. Keightley^{1*}

1 Institute of Evolutionary Biology, School of Biological Sciences, University of Edinburgh, Edinburgh, United Kingdom, 2 Department of Biology, William G. Davis Building, University of Toronto, Mississauga, Canada

¤ Current address: Institute of Plant Breeding, Seed Science and Population Genetics, University of Hohenheim, Stuttgart, Germany

* peter.keightley@ed.ac.uk

Abstract

The distribution of fitness effects (DFE) for new mutations is fundamental for many aspects of population and quantitative genetics. In this study, we have inferred the DFE in the singlecelled alga Chlamydomonas reinhardtii by estimating changes in the frequencies of 254 spontaneous mutations under experimental evolution and equating the frequency changes of linked mutations with their selection coefficients. We generated seven populations of recombinant haplotypes by crossing seven independently derived mutation accumulation lines carrying an average of 36 mutations in the haploid state to a mutation-free strain of the same genotype. We then allowed the populations to evolve under natural selection in the laboratory by serial transfer in liquid culture. We observed substantial and repeatable changes in the freguencies of many groups of linked mutations, and, surprisingly, as many mutations were observed to increase as decrease in frequency. Mutation frequencies were highly repeatable among replicates, suggesting that selection was the cause of the observed allele frequency changes. We developed a Bayesian Monte Carlo Markov Chain method to infer the DFE. This computes the likelihood of the observed distribution of changes of frequency, and obtains the posterior distribution of the selective effects of individual mutations, while assuming a two-sided gamma distribution of effects. We infer that the DFE is a highly leptokurtic distribution, and that approximately equal proportions of mutations have positive and negative effects on fitness. This result is consistent with what we have observed in previous work on a different C. reinhardtii strain, and suggests that a high fraction of new spontaneously arisen mutations are advantageous in a simple laboratory environment.

Author summary

Mutations are the ultimate source of genetic variation, form the raw material for evolution under natural selection, and generate a genetic load of harmful variants that are selectively

design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

removed from populations. Here, we have estimated the relative numbers of mutations with different sizes of effects on fitness in a laboratory strain of the model unicellular green alga *Chlamydomonas reinhardtii*. We estimated the fitness effects of mutations by measuring their frequency changes under experimental evolution in replicated populations where mutant allele frequencies started at 0.5 and where different genotypes might express different fitnesses. We developed a method to infer fitness effects based on changes of allele frequency; for example mutations that consistently decreased in frequency would be inferred to have a negative effect. While the majority of mutational effects were close to zero, we found that a high proportion of mutations increased in frequency under experimental evolution, implying that they had positive fitness effects in the laboratory environment. This is unexpected, because many mutations in natural populations have deleterious effects, and advantageous mutations appear to be rare.

Introduction

Understanding the nature of genetic variation for fitness requires an understanding of the origin of that variation from mutation. The distribution of fitness effects of mutations (DFE) describes the frequencies of mutations with differing magnitudes of effects, and is fundamental for many topics in evolutionary genetics, including the maintenance of genetic variation, the nature of genetic variation for quantitative traits and the genetic basis of adaptive evolution. The DFE specifies the relative frequencies of advantageous and deleterious mutations and the contributions of mutations with small and large effect sizes to fitness change and genetic variation. The DFE appears, for example, in the nearly neutral model of molecular evolution [1], which posits that patterns of molecular variation and between species change can be explained by mutations that have fitness effects close to $1/N_e$ (where N_e is the effective population size); this is a very small fitness effect for species with typically large N_e . The threat to population survival posed by mutation accumulation also depends on the nature of the DFE.

The DFE can be inferred experimentally or by statistical analysis of the frequencies of nucleotide variants at polymorphic sites [2]. In the latter approach, the DFE for deleterious amino acid-changing mutations can be estimated by analysis of the site frequency spectrum (SFS) for nonsynonymous variants under the assumption that their distribution of effects follows a prespecified distribution, such as a gamma distribution. When the SFS is combined with divergence data from another species the frequency and effects of advantageous nonsynonymous mutations can also be estimated [3,4]. Analysis of the SFS has been applied to genomic data from a wide range of taxonomic groups. Estimated DFEs for deleterious mutations are invariably strongly leptokurtic (L-shaped), the shape of the distribution varying between taxonomic groups [5], and there is usually a strong nearly neutral component. Analysis of the unfolded SFS suggests that at most a few percent of mutations are advantageous [6]. Inferring the DFE using standing variation within a population is relevant to the fitness effects of mutations in nature, but does not capture strongly positively or strongly negatively selected mutations, because these tend not contribute to standing nucleotide variation.

Inference of the DFE via experimental manipulation can be applied to mutations engineered in specific genes or at random genomic locations. One approach, which has similarities to the one described here, estimates the selection coefficients of induced mutations by tracking their frequency change over time under experimental evolution. This was pioneered by McDonald et al [7], who used deep sequencing to measure frequency changes of newly arisen mutations under experimental evolution in *Saccharomyces cerevisiae*. These frequency changes were used to estimate mutational effect sizes for fitness under adaptation, although McDonald et al [7] did not infer the full DFE. More recently, Flynn et al [8] used deep mutational scanning of yeast strains to infer the DFE for mutations in the Hsp90 gene. By competition between strains followed by deep sequencing, Flynn et al [8] quantified growth effects of many single codon changes encoding amino acid variants under standard environmental conditions and under five stress conditions. In standard conditions, the DFE is a leptokurtic distribution, containing a small proportion of beneficial mutations. However, the proportion of beneficial mutations was substantially higher in non-standard or stressful environments, especially high temperature and diamide.

The DFE can also be inferred for mutations induced at random locations in the genome. For example, Johnson et al [9] estimated the DFE for transposable element mediated insertion mutations in yeast by measuring mutation frequency changes under adaptation over time. Notably, lines with the highest initial fitness appeared to suffer the greatest fitness consequences from *de novo* insertion mutations.

Here, we study the spectrum of spontaneous mutations that occur at random locations across the entire genome. This is more relevant for several questions in evolutionary genetics, such as the maintenance of variation and the rate of decline of mean fitness from mutation accumulation, than previous studies that examine the fitness effects of random mutations in specific genes (e.g. [8]). Previously, we have attempted to infer the DFE for spontaneous mutations in the single-celled green alga Chlamydomonas reinhardtii using growth rate as a fitness measure [10]. We crossed mutation accumulation (MA) lines of the CC-2931 strain that had randomly accumulated spontaneous mutations for many generations with a mutation-free ancestral strain. We then measured growth rate and determined the genotypes of many recombinant lines carrying random combinations of mutations. We developed a Bayesian MCMC approach to estimate parameters of a DFE, which also enabled us to extract the estimated effects of individual mutations. This suggested a highly leptokurtic DFE, with a surprisingly high proportion of mutations (about 50%) increasing growth rate. Our previous results were unexpected, and in the present study we attempt to assess their generality with a somewhat different and potentially better approach, which infers selection on the basis of allele frequency change rather than differences in growth rate.

Here, we study MA lines of the haploid, unicellular green alga, *C. reinhardtii*, of a different strain (CC-2344) to that studied by [10]. We crossed lines that had undergone approximately 1,000 generations of spontaneous mutation accumulation with a mutation-free ancestral strain in order to generate many recombinants with different combinations of mutations. Rather than assaying individual recombinants, as in our previous experiment, we allow pools of recombinant haplotypes to compete against one another in an experimental evolution setting and measure changes of mutation frequency by deep sequencing. We develop a new MCMC approach to infer the DFE based on changes in frequency of linked mutations. Using this new experimental approach and a different strain, we infer that a surprisingly high proportion of mutations increase fitness in the standard laboratory environment.

Materials and methods

Mutation accumulation lines and compatible ancestor

We studied seven *C. reinhardtii* MA lines (L06, L09, L10, L12, L13, L14, L15) derived from strain CC-2344 (isolated in Pennsylvania, USA, in 1988) produced as described in [11] and sequenced as described in [12]. Because the MA lines and their ancestral strain are of the same mating type (mt+) and will not mate with one another, we first produced a "compatible ancestor" to which the MA lines could be crossed. This was done by backcrossing CC-2344 to a mt-strain (CC-1691) for 16 generations with the aim of producing a strain nearly identical to CC-

2344, with the exception of the region surrounding the mating type locus on chromosome 6 and the mitochondrial DNA. To confirm that the genetic composition of the compatible ancestor was as expected, we conducted whole genome sequencing as described in [12], mapped the reads together with publicly available data from the parental strains CC-2344 [13] and CC-1691 [14] to the *C. reinhardtii* reference genome (strain CC-503; version 5; [15]) and performed SNP calling as described in [16]. Sliding window analysis of genetic differences showed that this was accomplished successfully (S1 Fig): the compatible ancestor is genetically identical to the ancestral strain of the MA lines, CC-2344, with the exception of the mating type locus region at the distal part of chromosome 6, where it is genetically identical to the mt- donor strain CC-1691. Furthermore, the chloroplast DNA is identical to CC-2344 (mt+) and the mitochondrial DNA to CC-1691 (mt-), reflecting the uniparental mode of inheritance of the organelles.

Generation of populations of recombinants

To produce the starting populations of the experimental evolution experiments (designated t_0), we generated recombinant populations by mating each MA line with the compatible ancestor. First, we grew each MA line in Bold's medium [17] under standard conditions (23°C, 60% humidity, constant white light illumination) while shaking at 180 rpm to obtain a culture of 30 ml. Three lines had poor growth (L09, L12, and L15), so this procedure was done twice in order to obtain sufficient cell material. After three days, cultures were transferred to 50 ml falcon tubes, centrifuged at 3250 g for 5 minutes and the supernatant removed. Nitrogen-free conditions are required to trigger mating in *C. reinhardtii* [18], so we washed each cell pellet with 30 ml nitrogen-free Bold's medium, mixed, centrifuged at 3250 g for 5 minutes and removed the supernatant. We then resuspended the washed cell pellet in 45 ml nitrogen-free Bold's medium. The same procedure was done for the compatible ancestor.

To carry out the matings, we mixed 15 ml of the resuspended MA line cell culture with 15 ml of the resuspended compatible ancestor cell culture in a 50 ml falcon tube. Although the total cell number within the mating cultures was not directly estimated, cell densities of seven day cultures for C. reinhardtii CC-2344 MA lines previously measured ranged between 1.6-3.9 $x 10^5$ cells/ml. Therefore, the total number of cells in the 30 ml mating cultures can be conservatively estimated at 4.5×10^6 . Mixtures were incubated under standard growth conditions for 7 days in a slanting position (in order to increase the surface area) until zygote mats had formed at the surface. The remaining 30 ml of each culture served as control to allow the detection of mating failures (see below). Zygote mats were then transferred to a fresh 50 ml falcon tube containing 30 ml nitrogen-free Bold's medium and incubated in the dark under standard growth conditions for 5 days to allow the zygotes to mature. To kill any vegetative cells still associated with the zygote mats, we froze the zygote cultures and kept them at -20°C for 5 hours. After thawing at room temperature for approximately 1 hour, we transferred the zygote cultures to 500 ml conical flasks and added 30 ml of Bold's medium containing twice the standard concentration of nitrogen and 60 ml of standard Bold's medium to obtain a total volume of 120 ml with standard nitrogen concentration. The flask was then incubated under standard growth conditions while shaking at 250 rpm until zygotes had germinated (Fig 1A). This germination cell culture was then used to start the experimental evolution experiment. The control cultures were incubated as described for the zygote cultures and if any growth was visible after the freezing, the respective zygote culture would have been discarded and the whole procedure repeated, but no such instances were observed.

The number of zygotes that formed during the matings and that successfully germinated was not directly estimated, but optical density (OD) at 600nm was measured as a proxy for biomass two or three times for each MA line cross in the germination culture before t_0 . For the



Fig 1. Schematic overview of the experimental procedure. A) Matings. Cell cultures of the MA line and the compatible ancestor were mixed in a falcon tube and incubated until zygote mats had formed (1). The zygote mats were transferred to new falcon tubes, incubated in the dark to allow zygote maturation and then frozen at -20°C to kill off any vegatative cells (2). The culture with the matured zygote mat was transferred to a conical flask and incubated until zygotes germinated (3). B) Serial transfers. The germination culture was grown up until it was dense enough to start the three replicates and have enough cell material for DNA extraction. The three replicates were started with 1.2 ml of the original germination culture (4). Every three to four days 1.2 ml were transferred to fresh conical flasks and 2 ml of the remaining culture was used to measure OD (5). After nine transfers the end of the experiment was reached and the cells were collected for DNA extraction (6).

https://doi.org/10.1371/journal.pgen.1009840.g001

germination cultures in which these OD measurements captured the exponential growth phase, we were able to estimate the number of generations that occurred between the measurements. Assuming a constant rate of growth, these generation estimates could then be extrapolated across the entire growth period to estimate the total number of generations that occurred in the zygote cultures up to t_0 . This was possible for four of the seven germination cultures (L01, L06, L09 and L10) in which 4.6–5.1 generations were estimated to occur over time periods ranging between 12 to 23 days after zygotes were allowed to germinate up to t_0 . Furthermore, the final population size for each germination culture could be estimated using OD measurements converted to cell density using a standard curve for C. reinhardtii, including data from the CC-2344 ancestor, the compatible ancestor and the MA lines. Together, these values allowed us to estimate the initial population sizes of successfully germinated zygotes based on $\log N_0 = g \log 2 - \log N_t$, where g is the estimated number of generations that occurred during zygote culture growth up to t_0 and N_t is the estimated population size at the end of germination culture growth. These estimates ranged between 7,900 (L10) and 22,000 (L06) germinating zygotes, which are sufficiently large initial population sizes to avoid substantial stochastic shifts in allele frequencies early in the experiment.

Experimental evolution

We grew three replicate cell cultures from each MA line x compatible ancestor cross until approximately 60 generations had been reached (Fig 1B). We designate this time point t_t . For L06 and L09 this took 42 days and for L10, L12, L13, L14, and L15 this took 46 days. Each replicate was grown in a volume of 120 ml in a 250 ml conical flask under standard growth conditions while shaking at 250 rpm. We started each replicate with 1.2 ml of the germination cell culture and 118.8 ml Bold's medium. By using standard Bold's medium we ensured that the recombinants could not mate and grew entirely vegetatively. The remaining cells from the germination cell cultures were collected and frozen at -70°C for sequencing. We then transferred 1.2 ml of each culture to a fresh flask containing 118.8 ml Bold's medium on a three or four day cycle, or in the case of transfers 3, 6 and 9, over seven days in order to maximise biomass for DNA extraction and sequencing. In order to determine the number of serial transfers at which approximately 60 generations had been reached, OD at 600 nm of the cultures was measured at the end of each transfer growth period. We then calculated the number of generations that occurred within each serial transfer as follows: $g = (\log N_t - \log N_0) / \log 2$, where N_t is the measured OD of the culture at the end of the growth period. After nine transfers (approximately 60 generations) we collected cells for sequencing and froze the pellets as described for the time point 0 samples.

Sequencing and sequence data processing

Genomic DNA was obtained from time point 0 (t_0) and from each of the three t_t replicates for each MA line recombinant population by phenol-chloroform extraction [19]. DNA samples were sequenced on an Illumina Hiseq4000 platform by BGI Hong Kong with 150 bp pairedend reads to an average sequencing depth of 520.7x and 221.6x for time 0 and per time t replicate, respectively. Fastq reads were mapped to the *C. reinhardtii* reference genome (strain CC-503; version 5; [15]) with bwa-mem [20] and duplicate reads were removed with MarkDuplicates using picard tools. The four bam files of each MA line (time 0, and the three replicates of time t) were merged with samtools [21] into a single bam file. The 1,000-bp regions surrounding the locations of the mutations of interest (500 bp before and 500 bp after each mutation) were then realigned with HaplotypeCaller of GATK [22,23] to allow more accurate mapping of the mutations and more accurate allele frequency estimation. The realigned bam files were then split into the individual samples with SplitSamFile of GATK [22,23]. Samtools was then used to create pileup files from the realigned bam files using the mpileup command. Allele frequencies for the mutations of interest were then calculated with custom Perl scripts.

Mutations were classified as noncoding, synonymous or nonsynonymous relative to the v5.3 reference genome annotation. To assess the functional effects of mutations, SnpEff [24] was run using the pre-build *C. reinhardtii* annotation and with default parameters.

It has recently been suggested that the v5 reference genome contains some misassemblies [25,26]. Since misassemblies could introduce incorrect linkage relationships between mutations, we lifted over mutation coordinates to the highly contiguous Nanopore-based assembly of the strain CC-1690 [27], which is identical-by-descent at >95% of its genome with the original reference strain [14]. The v5 and CC-1690 assemblies were aligned with Cactus [28] with divergence between the genomes arbitrarily set at 0.004 and otherwise default parameters. Lift-over was then achieved from the resulting alignment using halLiftover [29]. Lifted over coordinates were used for the MCMC analysis described below.

Repeatability of mutation frequency between replicates

We estimated the repeatability of mutation frequency among the three replicate populations by partitioning the total variation in allele frequencies into three components: the variance in frequency among different MA line x compatible ancestor crosses (V_{MA}), the variance in frequency among mutations (V_M) within MA line x compatible ancestor crosses and the residual variation due to differences among the three replicate measures for each mutation (V_E). Variance components were estimated using the Lmer function in R, and repeatability was calculated as $V_M/(V_M + V_E)$.

C. reinhardtii genetic map

A genetic map is required in the MCMC analysis described below, which calculates frequency changes of linked mutations. We assume an overall genome-wide average rate of recombination, obtained from two published crosses between *C. reinhardtii* strains CC-2935 x CC-2936 and CC-408 x CC-2936, which together provide an estimate of 1cM per 87,000 bases [30]. A third cross [30] (CC-124 x CC-1010) that has an approximately 10-fold lower marker density was not included in our calculations.

Our analysis required the location of the individual mutations on a genetic map. Liu et al.'s [30] study does not provide sufficient resolution for this, but higher resolution estimates of the rate of recombination are available from a study of linkage disequilibrium in natural populations [31]. We used these estimates to adjust for variation in the rate of recombination among chromosomes, i.e. assumed a uniform rate of recombination per chromosome. Longer chromosomes have lower recombination rates, presumably due to a requirement for a minimum of 1 chiasmata per chromosome per meiosis (S2 Fig). Currently, the genetic map of *C. reinhardtii* is not sufficiently accurate to allow a detailed map to be directly included in the analysis. Note that the average number of mutations per chromosome per MA line is 2.1, so the majority of mutations will have behaved quasi-independently.

We used estimates of the population scaled recombination rate from [31] to adjust the rate of recombination estimated in [30] crossing experiment in order that the inverse rate of recombination (y_i , bp/cM) increases linearly by 0.000616 per 1 base pair increase in chromosome length (x_i) (S2 Fig), while keeping the overall average recombination unchanged, i.e.,

$$y_i = 0.000616x_i + k, (1)$$

where

$$k = 87,000 - 0.00616 \frac{\sum x_i^2}{\sum x_i},\tag{2}$$

(see S1 Text).

Computation of expected allele frequencies after experimental evolution

In this section we describe the computation of the expected frequencies of the mutations after one generation of recombination followed by t generations of experimental evolution, which are used in likelihood calculations and Bayesian inference. We assume that allele frequencies of each mutation are 0.5 after crosses between MA lines and their ancestor, and that changes of allele frequency then occur deterministically and independently among chromosomes, but that genetic linkage of mutations on the same chromosome leads to non-independent allele frequency changes under selection. Based on the assumed genetic map (see above), we first computed the expected frequencies of the n possible haplotypes generated by one round of meiosis involving an ancestral chromosome and a chromosome from a MA line in the absence of selection. In the model of experimental evolution following the cross, mutations have selection coefficients, s_{j_i} from which the overall fitness (w_i) of haplotype i carrying m_i mutations can be computed under multiplicative selection:

$$w_i = \prod_{j=1}^{m_i} (1 + \delta_j s_j), \tag{3}$$

Parameter	Definition
s _j	Selective effect of mutation <i>j</i> .
δ_{jk}	Deviation of the mean frequency of mutation j replicate k about its expectation.
α	Vector of scale parameters of the distribution of effects of mutations, elements 0 and 1 are for positive and negative effects, respectively.
β	Vector of shape parameters of the distribution of effects of mutations.
9	Frequency of positive effect mutations.

Table 1. Parameters of the model.

https://doi.org/10.1371/journal.pgen.1009840.t001

where δ_j takes the value 1 or 0 if the haplotype carries the mutant or wild type allele, respectively, for mutation *j*. The selection coefficients (*s*) are parameters of the model whose values are changed during MCMC runs. These and other such parameters are listed in Table 1.

Let the frequency of haplotype *i* at generation $t = \pi_{i,t}$. We then calculated the expected frequency of each haplotype after *t* generations of natural selection by iterating Eq (4) for *t* generations:

$$\pi_{i,t+1} = \pi_{i,t} w_i / w_t, \tag{4}$$

where w_t is the mean fitness of the *n* haplotypes at generation *t*. It is then straightforward to compute the expected frequencies of the individual mutations (*p*) at generation *t*.

Computation of likelihood

The log likelihood was the sum of three terms, the first involving the mutation frequencies after selection (p), the second involving mutation effects (s), which were assumed to be gamma distributed, and the third involving the total number of negative effect mutations (n_0) *versus* the total number of positive effect mutations (n_1), which were assumed to be sampled from a binomial distribution with mean q (the frequency of positive effect mutations, a parameter of the model, Table 1). Likelihood was computed assuming independence among the c chromosomes:

$$\log L = \sum_{i=1}^{c} \{ \log L(p)_i + \log L(s)_i \} + binomial(n_1, n_0 + n_1, q).$$
(5)

The log likelihood for the term involving the mutation frequencies on a chromosome (log *L* $(p)_i$) was computed assuming that allele frequencies of each of the *r* experimental evolution replicates are sampled from a log-normal distribution with mean p_j and standard deviation σ_{δ} , and that numbers of mutant and wild type reads for a replicate are sampled from a binomial distribution with mean $p_i + \delta_{ik}$:

$$\log L(p)_{i} = \sum_{j=1}^{m} \sum_{k=1}^{r} \{\log lognormal(p_{j} + \delta_{jk}, p_{j}, \sigma_{\delta} + \log binomial(x_{jk}, d_{jk}, p_{j} + \delta_{jk})\}, \quad (6)$$

where δ_{jk} are variables in the model that allow the allele frequency for each replicate (*k*) to be different from the overall expected frequency under selection for that mutation, and x_{jk} and d_{jk} are the numbers of mutant reads and the sequencing depth, respectively, for mutation *j* replicate *k*.

The log likelihood for the term involving the distribution of fitness effects of mutations (log $L(s)_i$) on a chromosome was computed assuming that the fitness effects are drawn from gamma distributions, which can have different parameters for positive and negative effect

mutations:

$$logL(s)_{i} = \sum_{j=1}^{m} log \ gamma(y_{j}s_{j}, \alpha_{\delta_{j}}, \beta_{\delta_{j}}), \tag{7}$$

where gamma() is the gamma probability density function (PDF), y_j takes the value -1 or 1 if mutation *j*'s fitness effect is negative or positive, respectively, δ_j is an indexing variable that takes the value 0 or 1 if mutation *j*'s fitness effect is negative or positive, respectively, and α_{δ_j} and β_{δ_j} are elements of vectors $\boldsymbol{\alpha}$ and $\boldsymbol{\beta}$, both of dimension two, containing the scale and shape parameters, respectively, of the gamma distributions of fitness effects. Elements 0 and 1 of $\boldsymbol{\alpha}$ and $\boldsymbol{\beta}$ contain parameters for negative and positive effect mutations, respectively.

Priors

These were designed to be uninformative. The prior for fitness effects of mutations was a uniform distribution bounded by -1 and +1. The prior for the frequency of positive-effect mutations (*q*) was uniform in the range 0 and 1. Priors for the shape and scale parameters of the gamma distribution of effects of mutations and the parameters related to the log normal distribution (σ_{δ} and δ_{jk}) were uniform in the range 0 to very large values.

MCMC implementation

We used the Metropolis Hastings algorithm to sample from the posterior distributions of the parameters (Table 1), based on the product of the log likelihood of the data and priors (which were designed to be uninformative, see above). Briefly, there was a burn-in of 10^9 iterations followed by sampling every 10^5 iterations up to iteration 10^{10} . Proposal deviates were sampled from normal distributions and added to the current parameter values. During the burn-in, the variance of a proposal distribution was either increased or decreased by a factor of 1.2 each iteration so that the average proportion of accepted proposals for each parameter was about 0.234, following [32]. The mode of the posterior distribution was used as the parameter estimate and 95% credible intervals were computed based on ranked posterior values.

Model comparisons

Model comparison was carried out using the Bayesian Information Criterion, BIC [33]: BIC = $k \log(n) - 2\log(L)$, where k is the number of parameters estimated in the model, n is the number of observations and L is the maximum likelihood for the model, computed as the modal log likelihood.

Results

Distribution of initial mutation frequency before experimental evolution

We crossed seven *C. reinhardtii* MA lines that had been independently derived from strain CC-2344 to a compatible ancestor of the same genetic background in order to generate populations of recombinants. In the generation following the cross, mutations are therefore expected to be at a frequency of 0.5. Recombinant haplotypes were then allowed to compete with one another in the absence of further recombination in standard laboratory conditions over the course of nine serial transfers. We sequenced samples from each MA line cross at the start (designated time 0) and end (designated time t) of experimental evolution in order to quantify changes in mutation frequency.

We identified 254 mutations in the seven MA lines, comprising 232 SNPs, 13 insertions and nine deletions (Table 2). Recombinant populations were sequenced at a high enough

		Mutat	ion type		Sequencing depth		
Line	SNP	INS	DEL	total	Time 0	Time t	
L06	21	2	2	25	628.0 (262.5; 47, 1366)	254.9 (117.2; 19, 665)	
L09	13	0	2	15	353.6 (154.5; 100, 536)	166.6 (84.1; 30, 306)	
L10	45	2	0	47	513.2 (226.8; 41, 1252)	216.4 (101.6; 10, 608)	
L12	50	2	0	52	509.6 (253.0; 8, 1767)	204.9 (134.6; 5, 1016)	
L13	35	4	2	41	487.1 (170.4; 120, 1011)	227.4 (85.6; 40, 526)	
L14	33	1	2	36	580.3 (214.1; 94, 1250)	263.6 (130.4; 23, 723)	
L15	35	2	1	38	520.3 (177.1; 92, 1134)	204.5 (71.9; 21, 437)	
all	232	13	9	254	520.7 (221.7; 8, 1767)	221.6 (111.0; 5, 1016)	

Table 2. Numbers of mutations in each MA line and sequencing depth statistics at the start and end of experimental evolution for the corresponding recombinant populations. Average sequencing depth across all mutations and replicates is shown along with the standard deviation and range in parenthesis.

https://doi.org/10.1371/journal.pgen.1009840.t002

depth at t_0 and t_t to allow accurate frequency estimation (Table 2). Sequencing depth varied among the mutations, but we did not observe a significant correlation between sequencing depth and mutation frequency (S3 Fig; time t_0 : r = 0.0741, P = 0.236 and t_i : r = 0.0244, P = 0.498; r = Spearman's correlation). Therefore, we can discount any effect of sequencing depth on mutation frequency.

The average mutation frequency at time 0 (p_0) was 0.481, which is close to the expected value of 0.5. Initial frequency of the different mutations showed considerable scatter, however, since the standard deviation was 0.141, and there were also noticeable differences in the distribution of frequencies between the recombinant populations. Lines L12 and L13, for example, had a relatively broad range of initial frequencies centering around 0.5, whereas L09 and L10 had a narrower distribution, with a mean close to 0.5 and a few mutations with very low frequencies (Fig 2). Unexpectedly, in the majority of lines there were mutations at frequencies close to zero at time 0, and additionally the frequency of one mutation in L15 was close to 1 at time 0 (S4 Fig). These extreme frequencies could be explained by natural selection changing mutation frequency in the generations of growth prior to the sequencing of the populations at time 0. This is corroborated by the presence of groups of linked mutations having similar frequencies at time 0 (S4 Fig).



Fig 2. Mutation frequencies at the start (p_0) and the end (p_t) of experimental evolution of the seven recombinant populations. Mutational types are indicated with symbols: open circles—SNPs, closed diamonds—Indels.

https://doi.org/10.1371/journal.pgen.1009840.g002

Mutations with extreme initial mean frequencies (e.g., $p_0 < 0.4$ or $p_0 > 0.6$) are not more frequently associated with specific annotations compared to mutations with non-extreme frequencies ($0.4 < p_0 < 0.6$); e.g., exonic *versus* non-exonic: P = 0.99 (bootstrap test).

Mutation frequencies after experimental evolution

There were substantial changes in the frequencies of many mutations after experimental evolution (p_t), but the magnitude and direction varied substantially among mutations (Fig 2). Mutation frequency was highly repeatable between the three replicates (r = 0.96), indicating that selection was the causal agent of frequency change.

The correspondence between mutation frequencies at the start and end of experimental evolution is shown in Fig 3. As previously mentioned, t_0 frequencies are centred around 0.5 (see also Figs 2 and S4), but in some cases t_0 frequencies were substantially different from 0.5. In the majority of these cases, frequency usually became even more extreme in the same direction by time *t*. There are exceptions, however, the most extreme of which correspond to points in the top-left and bottom-right quadrants of Fig 3. These are cases, comprising about 5% of mutations, where the allele frequency first moved down (top-left quadrant) and then moved up or *vice versa*. A possible explanation for this behaviour is a change in the direction of



Fig 3. Mutation frequencies at the start of experimental evolution (p_0) versus average mutation frequencies at the end of experimental evolution (p_t). The different recombinant populations are indicated with different colours and symbols. The dotted lines represent the frequencies of 0.5.

https://doi.org/10.1371/journal.pgen.1009840.g003

selection, which would imply that the environmental conditions before and after sampling for time 0 had changed. Another possibility is epistasis between a mutation and the genetic background of the mating type locus (which differed between donor and MA line strains). This could lead, for example, to purging of certain combinations of alleles and a consequent reduction in allele frequency at time 0.

Notwithstanding the unexpected cases mentioned above, there are other patterns apparent in the raw allele frequencies (S4 Fig). To account for allele frequency variation among replicates (which are mostly very consistent), in the subsequent analysis, we modelled variance among replicates of the same cross by assuming a lognormal model for the environmental variance between replicates of the same cross. Linked mutations usually moved in frequency in the same direction.

Estimation of the generation time

The number of generations over the course of experimental evolution were estimated based on OD measurements taken at each of the nine transfers. The mean number of generations across all recombinant populations was 59.8 with a standard deviation of 1.03. The estimated generation times were highly consistent between replicates (S5 Fig and S1 Table) and varied only slightly between recombinant populations from the different MA line x compatible ancestor crosses (S1 Table) where L06 had the lowest number of generations (58.1) and L15 the highest (60.7).

MCMC analysis to estimate the DFE

We proceeded to carry out the Bayesian MCMC analysis to estimate parameters of the DFE described in Materials and Methods (Table 3), assuming that the number of generations of natural selection under experimental evolution for each MA line recombinant population t = 60, fitting three different two-sided gamma distributions of fitness effects: a distribution with the same shape and scale parameters for positive- and negative-effect mutations, a distribution with different scale parameters and the same shape parameters for positive- and negative-effect mutations, and a distribution with different scale and shape parameters for positive- and negative-effect mutations. In each case, after a burn-in period, the sampler appeared to have converged (S6 Fig), and samples were drawn from the chain in order to obtain estimates of the posterior distributions of the various parameters.

Model comparison

We used the convention that if BIC(model A)–BIC(model B) < -10, there is strong evidence in favour of model A over model B [34]. The results (Table 3) therefore suggest that the most complex models (the model with different mean and shape parameters for positive- and negative-effect mutations) is marginally favoured over the simplest two-sided gamma mode, but the most complex model fits non-significantly better than the model with different means and the same shape.

Table 3. Models, their numbers of parameters related to the DFE and BIC values.

Model	Number of parameters	BIC
Two-sided gamma	2	1.6
Two-sided gamma, different means	3	7.2
Two-sided gamma, different means and shapes	4	12.3

https://doi.org/10.1371/journal.pgen.1009840.t003

	Mean mutati	on effect, β/α	Shape parameter, β			
Model	Negative	Positive	Negative	Positive	q	
Two-sided gamma	0.022		0.53		0.50	
	[0.019, 0.027]		[0.40, 0.69]		[0.43, 0.58]	
Two-sided gamma, different means	0.023	0.041	0.53		0.51	
	[0.017, 0.030]	[0.029, 0.060]	[0.40, 0.70]		[0.42, 0.58]	
Two-sided gamma, different means and shapes	0.023	0.023	0.53	0.69	0.49	
	[0.015, 0.032]	[0.015, 0.033]	[0.29, 1.0]	[0.31, 1.2]	[0.35, 0.64]	

Table 4. Models, and parameter estimates. 95% credible intervals are shown in square brackets.

https://doi.org/10.1371/journal.pgen.1009840.t004

Parameter estimates

We obtained estimates of parameter values based on the modes of the posterior distributions sampled from the MCMC chains. For the three different models evaluated, the parameter estimates are highly consistent (Table 4). The estimate of the proportion of positive-effect mutations (q) is close to 0.5 for each of the three models, and credible intervals are relatively narrow. This result is consistent with the observed bidirectional changes of mutation frequency. For each of the three models, estimates of the shape parameter of the distribution of effects are close to 0.5. This implies a highly leptokurtic distribution of fitness effect, in which the majority of effects cluster around zero, with a long tail of positive and negative effects. The estimated absolute average effect of a mutation is just over 2%, and the inferred DFE is shown in Fig 4.

Tests for differences between annotated mutations

Using the genome annotation for *C. reinhardtii*, we tested whether certain annotated types of mutations relating to protein-coding gene function have smaller or larger effects than others by calculating the difference in mean effect or mean squared effect between them (S2 Table). Based on bootstrap tests, all differences are nonsignificant.

As a complementary analysis, we estimated the effect of mutations on annotated coding sequences using SnpEff [24]. Of the ~42% of mutations that could be classified using this approach, 35% were estimated to have low impact and 61% moderate impact. These classifications largely coincided with the synonymous and nonsynonymous classifications used above. Only four high impact mutations were predicted, all of which were nonsense mutations.

Discussion

We have inferred the DFE for a set of spontaneous mutations from a MA experiment in *C*. *reinhardtii* by tracking their frequencies under natural selection in the laboratory and equating the observed frequency changes to the corresponding selection coefficients. The analysis of the data is made complicated by the fact that groups of mutations are linked on the same chromosome, implying that every mutation on a chromosome is expected to change in frequency even if only a subset of mutations is subject to selection. We have therefore crossed MA lines with an unmutated ancestor to obtain recombinants carrying the mutations in all possible combinations on each chromosome and measured frequency changes in the resulting recombinant populations after experimental evolution. We made use of the genetic map of *C. reinhardtii* for predicting changes of frequency of linked mutations on the same chromosome. In our data, frequency is estimated on the basis of numbers of sequencing reads, and we have used this information to compute the likelihood for a set of predicted frequencies. The likelihood is then used in a Bayesian setting to compute the posterior distribution of parameters of the DFE. To our knowledge, with the exception of our previous study [10], there have been no other



Selective effect

Fig 4. Inferred DFE, assuming the two-sided gamma distribution of effects with the same shape and scale parameters for negative- and positive-effect mutations.

https://doi.org/10.1371/journal.pgen.1009840.g004

attempts to fit a parameterized distribution to infer the DFE for new mutations. For example, Flynn et al [8] produced a graphical representation of the DFE based on estimates of individual mutation effects, but did not estimate the DFE's parameters within a statistical model. This is desirable, because a simple plot of the individual effects of mutations will be inflated by sampling variance, as will summary statistics derived from this distribution.

We observed substantial changes in the frequencies of the majority of mutations, and these changes were highly repeatable among replicates starting from the same base population. This is consistent with the action of natural selection under experimental evolution. The inferred DFE is broadly consistent with that inferred in our previous study on a different *C. reinhardtii* strain [10] in which we measured growth rate and assayed genotypes of many recombinants that emerged from a cross. If we assume a two-sided gamma DFE with equivalent scale and shape parameters for positive- and negative-effect mutations, the estimated proportions of positive-effect mutations are almost identical between the two studies. The estimated shape parameter in the current study is somewhat higher than the previous study ($\beta = 0.53$ versus 0.32), implying a somewhat less leptokurtic distribution, and the mean mutational effect is about four-fold higher (0.022 *versus* 0.0049). The biological significance of these differences is unknown, but our results show that the unexpectedly high proportion of positive-effect mutations found in our previous study [10] was not a unique finding.





https://doi.org/10.1371/journal.pgen.1009840.g005

Our estimated DFE is based on certain assumptions, and there are several potential causes of inaccuracy and/or bias. First, the mutations whose frequencies we tracked were those detected by Illumina sequencing in a previous study [12], but there are some mutations, including transposable element movements and large scale rearrangements, that we do not currently know about. Selection acting on these unknown mutations will therefore induce frequency changes at linked sites and generally inflate estimates of the strength of selection. Second, the linkage map for the strain we are working with may differ from the one that was assumed. Presumably any differences will lead to over/underestimation of effects, but not in a systematic way. Third, our analysis incorporates changes of frequency that occurred up to time 0 and changes that subsequently occurred up to time t, and the estimated selective effects are based on overall changes in frequency. We do not know the number of generations up to time 0, but it is clear that in several cases mutation frequencies had already changed by then. Furthermore we are assuming a certain value for the number of generations of experimental evolution, but do not have a precise measure of this. Fourth, selection coefficients are inferred using overall mutation frequency changes that occurred in the interval between the cross and

the sequencing. It is likely that different selection pressures acted in different phases of the experiment (i.e. generation of recombinant pools, experimental evolution by serial transfer), impacting frequency change of different mutation to varying extents.

We infer that there is a high frequency of mutations with positive effects on fitness, which we also observed in our previous study in a different *C. reinhardtii* strain [10]. Such a high frequency is a surprising finding, since it has long been argued that the majority of new mutations are likely to be neutral or deleterious (summarized by Keightley and Lynch [35]. Broadly, the majority of nucleotide sites in compact genomes such as in *C. reinhardtii* are at sites that are selectively constrained, and relatively few sites can evolve free from the influence of natural selection. In nature, the dominant force of natural selection appears to be purifying selection, since fitness in natural populations is likely to be close to an adaptive peak and most changes are therefore harmful. Direct evidence for this comes from attempts to infer the fraction of advantageous amino acid mutations, based on analysis of the site frequency spectrum (e.g., [4,6]). Most studies utilizing mutation accumulation also suggest that the net directional effect of mutations is negative (summarized in [36]).

One possible explanation for the higher than expected frequency of advantageous mutations is that there was selection against deleterious mutations and in favour of advantageous mutations during the generation of the MA lines. This could have affected the strain that was the subject of the current experiment and the strain studied in [10]. In the experiment described here, there were approximately 11 generations of growth between each transfer in the MA experiment, where selection could operate to change the frequencies of de novo mutations. To investigate the influence of selection on the estimated DFE, we implemented the method recently developed by Wahl and Agashe [37] to predict the extent of under- or over-contribution to the DFE for mutations with given selection coefficients, assuming that there is a doubling of cell number for t = 11 generations during mutation accumulation. Based on Eq (2) in [37], and assuming a two-sided gamma distribution with parameters specified in Table 4, the corrected estimate for the frequency of positive effect mutations is q = 0.46 (uncorrected = 0.50). This suggests that selection during mutation accumulation had only a modest impact. Presumably, this is because our inferred DFE is leptokurtic, and most of the density is concentrated near zero. Furthermore, the mean absolute selective effect is relatively low (0.022), and the 11 generations of growth between transfers in the MA experiment were insufficient to lead to substantial frequency changes over the bulk of the distribution. The contrast between corrected and uncorrected DFEs (Fig 5) shows only a slight downward shift in the corrected distribution.

Although our experimental design attempted to mitigate it, there will also have been selection against those colonies that were too small to see at the time of transfer during the MA experiment. This would most likely select against strongly deleterious mutations, but the effect of this between-colony selection is difficult to quantify.

If selection during mutation accumulation was not the explanation for the high frequency of positively selected mutations, then we must seek other explanations. Although the strains we studied were isolated three decades ago, *C. reinhardtii* can be maintained for long periods without cell division and it is probable that they have passed through insufficient generations to adapt to the novel laboratory environment. Fitness is therefore likely to be far from an adaptive peak, and therefore the fraction of advantageous mutations is expected to be higher than in a natural environment [38]. Little is known about the ecology of *C. reinhardtii* has been isolated from soil and may also be present in freshwater, where the species experiences day/night cycles, fluctuating temperatures and resource availability, biotic interactions (predators, pathogens etc.), and so on. Adaptations to life in the field include a complex metabolism (autotrophy and heterotrophy) and motility in response to light and nutrients [39]. Lit is plausible that loss

of function mutations in genes no longer required in the laboratory are advantageous. Loss of function mutations have frequently arisen under laboratory conditions, for example certain strains have lost the ability to utilise nitrate after culture with an alternative nitrogen source [14,40], although it is unknown if the underlying mutations in such cases were positively selected. If adaptation to the laboratory has contributed to the high proportion of beneficial mutations, this may have implications for further attempts to infer the distribution of fitness effects of new mutations using microbial laboratory models.

Supporting information

S1 Text. Calculation of scaled recombination rates per chromosome. (PDF)

S1 Table. Estimated numbers of generations of experimental evolution. (PDF)

S2 Table. Difference in effect sizes and effect sizes squared between annotation categories along with P-values obtained by bootstrapping by mutation 10,000 times. (PDF)

S1 Fig. Genetic differences (SNPs and small indels) along the 17 chromosomes and the organelle genomes between the compatible ancestor and its two parental strains, strain CC-2344 (mt+, red) and the mt- donor strain CC-1691 (black). The proportion of genetic differences for 80-kb windows in the case of the chromosomes and for 2-kb and 200-bp windows for the cpDNA and mtDNA, respectively, were calculated based on variant tables extracted from the VCF file using the VariantsToTable tool of GATK [22,23]. (TIF)

S2 Fig. Relationship between chromosome length (in bases) and base pairs per cM, estimated from pairwise linkage disequilibrium in *C. reinhardtii* from a natural population (data from [31[). (Pearson correlation r = 0.56; linear regression: bp/cM = 0.00616 x length + 43,900, P = 0.015).

(PNG)

S3 Fig. Mutation frequencies *versus* **sequencing depth at times 0 and** *t*. Mean sequencing depth for the mutations are 520.7x and 221.6x for times 0 and *t*, respectively. (TIF)

S4 Fig. Frequencies of the individual mutations before and after experimental evolution. Mutations of each MA line are shown from top to bottom in the order in which they occur in the genome. Squares denote the mutation frequencies at t_0 , and stars denote the mutation frequencies of the three replicates at t_t . The different MA line recombinant pools are shown in the different panels.

(TIF)

S5 Fig. Increase of generations during the course of the experiment. Each panel shows the cumulative number of generations of the three replicates of each of the seven recombinant populations derived from a backcross between MA line and compatible ancestor. Number of generations was estimated from OD measurements conducted at each transfer. (TIF)

S6 Fig. Output of MCMC sampler for three DFEs. (TIF)

Author Contributions

Conceptualization: Rob W. Ness, Nick Colegrave, Peter D. Keightley.

Data curation: Katharina B. Böndel.

Formal analysis: Katharina B. Böndel, Toby Samuels, Rory J. Craig, Nick Colegrave, Peter D. Keightley.

Funding acquisition: Rob W. Ness, Nick Colegrave, Peter D. Keightley.

Investigation: Katharina B. Böndel.

Methodology: Katharina B. Böndel, Peter D. Keightley.

Software: Peter D. Keightley.

Supervision: Peter D. Keightley.

Writing - original draft: Katharina B. Böndel, Peter D. Keightley.

Writing – review & editing: Toby Samuels, Rory J. Craig, Rob W. Ness, Nick Colegrave, Peter D. Keightley.

References

- Ohta T. Slightly deleterious mutant substitutions in evolution. Nature. 1973; 246: 96–98. <u>https://doi.org/10.1038/246096a0 PMID: 4585855</u>
- Eyre-Walker A, Keightley PD. The distribution of fitness effects of new mutations. Nat Rev Genet. 2007; 8: 610–618. https://doi.org/10.1038/nrg2146 PMID: 17637733
- Eyre-Walker A, Keightley PD. Estimating the rate of adaptive molecular evolution in the presence of slightly deleterious mutations and population size change. Mol Biol Evol. 2009; 26: 2097–2108. https://doi.org/10.1093/molbev/msp119 PMID: 19535738
- Tataru P, Mollion M, Glémin S, Bataillon T. Inference of distribution of fitness effects and proportion of adaptive substitutions from polymorphism data. Genetics. 2017; 207: 1103–1119. <u>https://doi.org/10. 1534/genetics.117.300323</u> PMID: 28951530
- 5. Chen J, Glémin S, Lascoux M. Genetic diversity and the efficacy of purifying selection across plant and animal species. Mol Biol Evol. 2017; https://doi.org/10.1093/molbev/msx088 PMID: 28333215
- Keightley PD, Campos JL, Booker TR, Charlesworth B. Inferring the frequency spectrum of derived variants to quantify adaptive molecular evolution in protein-coding genes of *Drosophila melanogaster*. Genetics. 2016; 203: 975–984. https://doi.org/10.1534/genetics.116.188102 PMID: 27098912
- McDonald MJ, Rice DP, Desai MM. Sex speeds adaptation by altering the dynamics of molecular evolution. Nature. 2016; 531: 233–236. https://doi.org/10.1038/nature17143 PMID: 26909573
- Flynn J, Rossouw A, Cote-Hammarlof P, Fragata I, Mavor D, Hollins C et al. Comprehensive fitness maps of Hsp90 show widespread environmental dependence. eLife. 2020; 9:e53810. <u>https://doi.org/ 10.7554/eLife.53810 PMID: 32129763</u>
- Johnson MS, Martsul A, Kryazhimskiy S, Desai MM. Higher-fitness yeast genotypes are less robust to deleterious mutations. Science. 2019; 366: 490–493. https://doi.org/10.1126/science.aay4199 PMID: 31649199
- Böndel KB, Kraemer SA, Samuels TS, McClean D, Lachapelle J, Ness RW et al. Inferring the distribution of fitness effects of spontaneous mutations in *Chlamydomonas reinhardtii*. PLoS Biol. 2019; 17: e3000192. https://doi.org/10.1371/journal.pbio.3000192 PMID: 31242179
- Morgan AD, Ness RW, Keightley PD, Colegrave N. Spontaneous mutation accumulation in multiple strains of the green alga, *Chlamydomonas reinhardtii*. Evolution. 2014; 68: 2589–2602. https://doi.org/ 10.1111/evo.12448 PMID: 24826801
- Ness RW, Morgan AD, Vasanthakrishnan RB, Colegrave N, Keightley PD. Extensive de novo mutation rate variation between individuals and across the genome of *Chlamydomonas reinhardtii*. Genome Res. 2015; 25: 1739–1749. https://doi.org/10.1101/gr.191494.115 PMID: 26260971
- Flowers JM, Hazzouri KM, Pham GM, Rosas U, Bahmani T, Khraiwesh B et al. Extensive natural variation in the model green alga *Chlamydomonas reinhardtii*. Plant Cell. 2015; 27: 2353–2369.

- Gallaher SD, Fitz-Gibbon ST, Glaesener AG, Pellegrini M, Merchant SS. Chlamydomonas genome resource for laboratory strains reveals a mosaic of sequence variation, identifies true strain histories, and enables strain-specific studies. Plant Cell. 2015; 27: 2335–2352. https://doi.org/10.1105/tpc.15. 00508 PMID: 26307380
- Blaby IK, Blaby-Haas CE, Tourasse N, Hom EF, Lopez D, Aksoy M et al. The *Chlamydomonas* genome project: a decade on. Trends Plant Sci. 2014; 19: 672–680. https://doi.org/10.1016/j.tplants.2014.05. 008 PMID: 24950814
- Craig RJ, Böndel KB, Arakawa K., Nakada T, Ito T, Bell G. et al. Patterns of population structure and complex haplotype sharing among field isolates of the green alga *Chlamydomonas reinhardtii*. Mol Ecol. 2019; 28: 3977–3993. https://doi.org/10.1111/mec.15193 PMID: 31338894
- 17. Bold HC. The cultivation of algae. Bot Rev. 1942; 8: 69–138.
- Sager R, Granick S. Nutritional control of sexuality in *Chlamydomonas reinhardtii*. J Gene Physiol. 1954; 37: 729–742.
- Ness RW, Morgan AD, Colegrave N, Keightley PD. Estimate of the Spontaneous Mutation Rate in *Chlamydomonas reinhardtii*. Genetics. 2012; 192: 1447–1454. https://doi.org/10.1534/genetics.112. 145078 PMID: 23051642
- Li H, Durbin R Fast and accurate short read alignment with Burrows–Wheeler transform. Bioinformatics. 2009; 25: 1754–1760. https://doi.org/10.1093/bioinformatics/btp324 PMID: 19451168
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25: 2078–2079. <u>https://doi.org/10.1093/bioinformatics/btp352</u> PMID: 19505943
- McKenna A, Hanna M, Banks E, Sivachenko A, Cibulskis K, Kernytsky A et al. The Genome Analysis Toolkit: a MapReduce framework for analyzing next-generation DNA sequencing data. Genome Res. 2010; 20: 1297–1303. https://doi.org/10.1101/gr.107524.110 PMID: 20644199
- 23. DePristo MA, Banks E, Poplin R, Garimella KV, Maguire JR, Hartl C et al. A framework for variation discovery and genotyping using next-generation DNA sequencing data. Nat Genet. 2011; 43: 491–498. https://doi.org/10.1038/ng.806 PMID: 21478889
- Cingolani P, Platts A, Wang LL, Coon M., Nguyen T, Wang LL et al. A program for annotating and predicting the effects of single nucleotide polymorphisms, SnpEff: SNPs in the genome of Drosophila melanogaster strain w1118; iso-2; iso-3. Fly (Austin). 2012; 6: 80–92. <u>https://doi.org/10.4161/fly.19695</u> PMID: 22728672
- Salomé PA, Merchant SS. A Series of fortunate events: Introducing *Chlamydomonas* as a reference organism. Plant Cell. 2019; 31: 1682–1707. https://doi.org/10.1105/tpc.18.00952 PMID: 31189738
- Craig RJ, Hasan AR, Ness RW, Keightley PD. Comparative genomics of *Chlamydomonas*. Plant Cell. 2021; 33: 1016–1041. https://doi.org/10.1093/plcell/koab026 PMID: 33793842
- O'Donnell S, Chaux F, Fischer G. Highly contiguous Nanopore genome assembly of *Chlamydomonas* reinhardtii CC-1690. Microbiol Resour Announc. 2020; 9: e00726–00720. <u>https://doi.org/10.1128/</u> MRA.00726-20 PMID: 32912911
- Armstrong J, Hickey G, Diekhans M, Fiddes IT, Novak AM Deran A, et al. Progressive Cactus is a multiple-genome aligner for the thousand-genome era. Nature. 2020; 587: 246–251. https://doi.org/10. 1038/s41586-020-2871-y PMID: 33177663
- Hickey G, Paten B, Earl D, Zerbino D, Haussler D. HAL: a hierarchical format for storing and analyzing multiple genome alignments. *Bioinformatics*. 2013; 29: 1341–1342. <u>https://doi.org/10.1093/bioinformatics/btt128</u> PMID: 23505295
- Liu H, Huang J, Sun X, Li J, Hu Y, Yu L, et al. (2018). Tetrad analysis in plants and fungi finds large differences in gene conversion rates but no GC bias. Nat Ecol Evol. 2: 164–173. https://doi.org/10.1038/ s41559-017-0372-7 PMID: 29158556
- Hasan AR, Ness, RW. Recombination rate variation and infrequent sex influence genetic diversity in *Chlamydomonas reinhardtii*. Genome Biol Evol. 2020; 12: 370–380. <u>https://doi.org/10.1093/gbe/</u> evaa057 PMID: 32181819
- 32. Gelman A, Roberts GO, Gilks WR. Efficient Metropolis jumping rules. Bayesian Statistics. 1996; 5: 599–608.
- 33. Schwarz GE. Estimating the dimension of a model. Ann Stat. 1978; 6: 461–464.
- 34. Raftery AE. Bayesian model selection in social research. Sociological Methodology 1995; 25: 111–163.
- **35.** Keightley PD, Lynch M. Towards a realistic model of mutations affecting fitness. Evolution. 2003; 57: 683–685. https://doi.org/10.1111/j.0014-3820.2003.tb01561.x PMID: 12703958
- Halligan DL, Keightley PD. Spontaneous mutation accumulation studies in evolutionary genetics. Annu Rev Ecol Evol Syst. 2009; 40: 151–172.

- Wahl LM, Agashe D. Selection bias in mutation accumulation. Evolution. 2022; <u>https://doi.org/10.1111/</u> evo.14430 PMID: 34989408
- Orr HA. The Population Genetics of Adaptation: The distribution of factors fixed during adaptive evolution. Evolution. 1998; 52: 935–949. https://doi.org/10.1111/j.1558-5646.1998.tb01823.x PMID: 28565213
- Sasso S, Stibor H, Mittag M, Grossman AR 2018. From molecular manipulation of domesticated *Chlamydomonas reinhardtii* to survival in nature. eLife. 2019; 7.
- **40.** Harris EH. The Chlamydomonas Sourcebook (Second Edition): Introduction to Chlamydomonas and Its laboratory use. Academic Press. 2009.